Author's response to reviews

Title: Targeting condom distribution at high risk places increases condom utilization - evidence from an intervention study in Livingstone, Zambia

Authors:

Ingvild Fossbard Sandøy (Ingvild.sandoy@cih.uib.no)
Cosmas Zyaambo (czyaambo@yahoo.com)
Charles Michelo (ccmichelo@yahoo.com)
Knut Fylkesnes (Knut.Fylkesnes@cih.uib.no)

Version: 7 Date: 2 January 2012

Author's response to reviews:

Dear editor,

The following changes have been done at this proof-reading stage:

- I have inserted the subheading "Conclusion" before the final paragraph
- I have removed reference 5 since it has not yet been published. I have referred to this manuscript in the text on p. 21. Please note that I had to change the numbering of the other references as a result, and that the changes may have affected the digital link between the references in the text and the reference list
- I have cleaned up the quotes (i.e. removed some unnecessary words)
- I have specified the themes used in the qualitative analysis in the Methods section
- In sentences where I had previously written that there were changes in the percentages "from x% to y%", you had changed this to "from x% - y%". I have now changed this to "x% to y%" and I hope this is acceptable. The way you had changed it looked as if the sentences described variance instead of changes.

Thank you very much for accepting this paper.

Yours sincerely,
Ingvild Fossbard Sandøy